Author's response to reviews

Title: Long-term cultivation of colorectal carcinoma cells with anti-cancer drugs induces drug resistance and telomere elongation: an in vitro study

Authors:
Noritsugu Kuranaga (grd1409@gr.ndmc.ac.jp)
Nariyoshi Shinomiya (shinomi@ndmc.ac.jp)
Hidetaka Mochizuki (mochizuki@me.ndmc.ac.jp)

Version: 7 Date: 30 Jul 2001

Dear Sir:

According to the reviewers'/editor's suggestion, I have improved the manuscript as follows:
1) I have rewritten the title of our manuscript that it includes the type of intervention, the condition or group, and the methodology.
2) The abstract has been shortened (249 words) and a methods section has been added.
3) The full address has been given for each author affiliation.
4) The references have been rewritten according to the BioMed Central style.
I think that the revision is satisfactory, and hope that the manuscript has been accepted for publication in its present form. Thank you very much for your cooperation.